Author's response to reviews

Title: Vegetarian diets are associated with healthy mood states: a cross-sectional study in Seventh Day Adventist adults

Authors:

Bonnie L Beezhold (bonnie.beezhold@asu.edu)
Carol S Johnston (carol.johnston@asu.edu)
Deanna R Daigle (deanna.daigle@asu.edu)

Version: 2 Date: 29 July 2009

Author's response to reviews:

July 28, 2009

Dr. Nehme Gabriel
Editor-in-Chief
Nutrition Journal

Dear Dr. Gabriel:

Please accept resubmission of our article entitled, “Vegetarian diets are associated with healthy mood states: a cross-sectional study in Seventh Day Adventist adults”, as a Short Report in your journal. Unfortunately, the format for Short Report was not followed correctly in our earlier submission, and we apologize for the oversight. We have revised our format to fit the body of the manuscript into one “Findings” section with no more than 1500 words. We trust that everything is now in order.

Sincerely,

Bonnie L. Beezhold, PhD, MHS
Carol S. Johnston, PhD, RD
Deanna R. Daigle, BS